%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%        This is a template file for                 %%%%%%%%%%%%%
%%%%%%%%%%%%%%%%   ''Chamchuri Journal of Mathematics (CJM)''       %%%%%%%%%%%%%
%%%%%%%%%%%%% Authors should submit papers according to this format %%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\documentclass[twoside,8pt,b5paper]{article}

\usepackage{graphicx,amsmath,latexsym,amssymb,amsthm,fancyhdr}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%% Your additional packages %%%%
% \usepackage{   }
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%% PAGE LAYOUT  *********  DO NOT MAKE ANY CHANGES HERE *********  %%%%%%%%%%%

\hoffset=0.0in
\voffset=-0.25in

\hoffset=0.0in
\voffset=-0.25in

\setlength\oddsidemargin{0.8in}
\setlength\evensidemargin{0.5in}
\setlength\topmargin{-0.3cm}
\setlength\headheight{0cm}
\setlength\headsep{25pt}
\setlength\topskip{0.0cm}

\setlength\textheight{18.0cm}
\setlength\textwidth{12.5cm}

\setlength\marginparsep{0cm}
\setlength\marginparwidth{0cm}
\setlength\footskip{1.0cm}

%%%%%%%%%%%%%%%%% END PAGE LAYOUT  %%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%% ENVIRONMENT  %%%%%%%%%%%%%%%%%%%%
%%%%%%% THEOREM - PROPOSITION - COROLLARY - %%%%%%%
%%%%%%% LEMMA - CONJECTURE Environments     %%%%%%%
%%%%%%% REMARK - CONVENTION - NOTATION Environments   %%%
%%%%%%% DEFINITION - AXIOM - EXAMPLE        %%%%%%%
%%%%%%% COUNTEREXAMPLE - NOTE - QUESTION Environments %%%

\newcounter{Theorem}

\theoremstyle{plain}
\newtheorem{thm}[Theorem]{\bf Theorem}
\newtheorem{prop}[Theorem]{\bf Proposition}
\newtheorem{cor}[Theorem]{\bf Corollary}
\newtheorem{lemma}[Theorem]{\bf Lemma}
\newtheorem{conj}[Theorem]{\bf Conjecture}

\theoremstyle{remark}
\newtheorem{remark}[Theorem]{\bf Remark}
\newtheorem{convention}[Theorem]{\bf Convention}
\newtheorem{notation}[Theorem]{\bf Notation}

\theoremstyle{definition}
\newtheorem{defn}[Theorem]{\bf Definition}
\newtheorem{axiom}[Theorem]{\bf Axiom}
\newtheorem{example}[Theorem]{\bf Example}
\newtheorem{counterexample}[Theorem]{\bf Counterexample}
\newtheorem{note}[Theorem]{\bf Note}
\newtheorem{question}[Theorem]{\bf Question}

\numberwithin{Theorem}{section}

\renewcommand\refname{References}

%%%%%%%%%%%%%%%%% END ENVIRONMENT  %%%%%%%%%%%%%%%%%%%%

%%%%%%%%%%%%%%  AUTHOR-COMMANDS  %%%%%%%%%%%%%%%%
%%% Examples of commands used in the article %%%%

\newcommand{\N}{\ensuremath{\mathbb{N}}}
\newcommand{\Z}{\ensuremath{\mathbb{Z}}}
\newcommand{\Q}{\ensuremath{\mathbb{Q}}}

\DeclareMathOperator{\Var}{Var}
\DeclareMathOperator{\Ima}{Im}

%%%%%%%%%%%%%%  END AUTHOR-COMMANDS  %%%%%%%%%%%%%%%%%%%%

\mathsurround=1pt

\renewcommand{\baselinestretch}{1.2}

\renewcommand{\headrulewidth}{0pt}
\pagestyle{fancy}
\fancyhead{} % clear all header fields
\fancyhead[RE]{\footnotesize{\thepage \hfill \textit{Chamchuri J.~Math. \textbf{5}(2013): N1.~Surname1, N2.~Surname2 and N3.~Surname3}}}
                %%% using N1. Surname1 instead of Name1 Surname1
\fancyhead[LO]{\footnotesize{\textit{Title of your article} \hfill \thepage}}
\fancyfoot{} % clear all footer fields

\begin{document}

\thispagestyle{empty}

\setcounter{page}{1}% ignore this

\vspace*{-1.5cm}

\noindent {\footnotesize{\textbf{Chamchuri Journal of Mathematics}}} \\
\noindent {\footnotesize{\textsc{Volume 5(2013), \pageref{1st-page}--\pageref{last-page}}}} \\
\noindent {\footnotesize{\texttt{http://www.math.sc.chula.ac.th/cjm}}}\\

%\vspace{-2.4cm}
%
%\begin{figure}[h]
%\begin{flushright}
%\includegraphics[width=2.5cm,height=1.5cm]{chamchuri.eps}
%\end{flushright}
%\end{figure}

\label{1st-page}

\renewcommand{\thefootnote}{\fnsymbol{footnote}}

\begin{center}
{\LARGE{\textbf{Title of your paper (Line 1)}}} \\[0.2cm]
{\LARGE{\textbf{Title of your paper (Line 2)}}} \\[0.4cm]
%
{\Large{Name1 Surname1{\footnote{\textit{The author is supported by xxx}}},
        Name2 Surname2{\footnote{\textit{Corresponding author}}}}} \\[0.2cm]
{\Large{and Name3 Surname3}} \\[0.2cm]
\end{center}

\vspace{0.25cm}

\begin{flushright}
\begin{small}
\textit{Received dd mm 2013} \\
\textit{Revised dd mm 2013} \\
\textit{Accepted dd mm 2013} \\
\end{small}
\end{flushright}

\renewcommand{\thefootnote}{\arabic{footnote}}

\bigskip

\noindent
\textbf{Abstract:\ } Your abstract.

\vspace{0.6cm}

\noindent
\textbf{Keywords:\ } xxxxx  % using a comma in order to separate your list of keywords, no period

\medskip

\noindent
\textbf{2000 Mathematics Subject Classification:\ } xxxxxxx % using a comma in order to separate the list, no period

\bigskip

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{The First Section}

%To cite refences listed in the bibliography use, for example, \cite{chen}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
The followings are various environment templates you might need
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\begin{defn}
fff
\end{defn}

We see from \cite{m49} that xxxxx

\begin{thm}
abc
\begin{equation}
x+y\leq 2
\end{equation}
\end{thm}

\begin{proof}
xxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxxx

\end{proof}

\begin{cor}
cdefg
\end{cor}

\begin{prop} \cite{chen}
cdefg
\end{prop}

\begin{lemma}
cdf
\end{lemma}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{The Second Section}

\begin{prop}
dfg
\end{prop}

Moreover, \cite{web1} suggests that xxxx

\begin{cor}
aaaa
\end{cor}

\begin{remark}
eee
\end{remark}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{The Third Section}

\begin{example}
From \cite{Matsumura}, xxx
\end{example}


%%%%%%%%%%%%%%%%%%%%%%%%%%
\bigskip

\noindent{\bf Acknowledgements:\ } xxx


%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{thebibliography}{99}

%%%% Please order the surnames of the first authors alphabetically


%Reference to a journal publication:

\bibitem{chen} L.H.Y.~Chen and Q-M.~Shao,  A Non-Uniform Berry-Esseen Bound via Stein's Method,
{\it Probab. Theory Relat.~ Fields}, {\bf 120}(2001), 236--254.


\bibitem{m49} K.~Mahler, On a theorem of Liouville in fields of positive characteristic, {\it Canadian J.~Math.},
{\bf 1}(1949), 397--400.


%Reference to a book:

\bibitem{Matsumura} H.~Matsumura,  \textit{Commutative ring theory}, (3rd ed.), Cambridge University Press, 1992.
{\it Probab.~Theory Relat.~Fields}, {\bf 120}(2001), 236--254.


%Reference to a web source:

\bibitem{web1} J.~Smith, One of Volvo's core values. [Online] Available: \\ \texttt{http://www.volvo.com/environment/index.htm},
(July 7, 1999).

\end{thebibliography}

\vspace{0.6cm}


\begin{small}
\noindent
Name1 Surname1 \\
Address-1st-line \\
Address-2nd-line \\
Address-3rd-line \\
Email: \texttt{author1@xxx.com}
\end{small}

\bigskip

\begin{small}
\noindent
Name2 Surname2 \\
Address-1st-line \\
Address-2nd-line \\
Address-3rd-line \\
Email: \texttt{author2@xxx.com}
\end{small}

\label{last-page}

\end{document}


